Author's response to reviews

Title: Systematic Review of topotecan (Hycamtin) in relapsed small cell lung cancer

Authors:

Rob Riemsma (rob@systematic-reviews.com)
Jean PFHA Simons (jp.simons@live.nl)
Zahid Bashir (zahid.2.bashir@gsk.com)
Caroline L Gooch (caroline.l.gooch@gsk.com)
Jos Kleijnen (jos@systematic-reviews.com)

Version: 2 Date: 5 November 2009

Author's response to reviews:

We have included the following 'Competing interests' section between the Conclusions and Authors' contributions as requested:

Competing interests

The project was funded by GlaxoSmithKline. Zahid Bashir and Caroline Gooch both are employed by GlaxoSmithKline, UK. Rob Riemsma, Jean Simons and Jos Kleijnen have no other conflicts of interest. Furthermore, RR, JS and JK had ultimate editorial control of the manuscript.